Title: A randomised, open-label study of insulin glargine or neutral protamine Hagedorn insulin in Chinese paediatric patients with type 1 diabetes mellitus

Authors:

M Liu (liumin2001@hotmail.com)
ZG Zhou (zhouzg@hotmail.com)
JH Yan (yanjh79@163.com)
P Lin (lipin21@126.com)
WH Song (13754881693@163.com)
JF Fu (ffj68@qq.com)
XB Chen (xiaobochen6362@163.com)
WG Zhao (zhaoweigang@263.net)
L Xi (drxili@163.com)
XP Luo (xpluo@tjh.tjmu.edu.cn)
L Sha (liang.sha@sanofi.com)
XY Deng (Rita.Deng@sanofi.com)
Chunxiu Gong (chunxiugong@163.com)

Version: 3 Date: 28 Sep 2016

Author’s response to reviews:

Reviewer reports:

Please note that the following revision is compulsory:

All manuscripts must contain the following sections under the heading 'Declarations':

Ethics approval and consent to participate
Response:

Dear Editor,

Thanks for pointing this out. As requested, we have added the Declarations section in the manuscript after the Abbreviations section (Page 12-13). The Declarations section includes the following subsections:

- Ethics approval and consent to participate
- Consent for publication
- Availability of data and material
- Competing interests
- Funding
- Authors’ contributions
- Acknowledgements

To avoid redundancy, we have deleted funding-related information from the Acknowledgements subsection (page 13) and we have provided detailed funding information under the Funding subsection (page 13).